Author's response to reviews

Title: NF-kappa B genes have a major role in Inflammatory Breast Cancer

Authors:

Florence Lerebours (f.lerebours@stcloud-huguenin.org)
Sophie Vacher (oncogenetique@stcloud-huguenin.org)
Catherine Andrieu (oncogenetique@stcloud-huguenin.org)
Marc Espie (marc.espie@sls.aphp.fr)
Michel Marty (m.marty@sls.aphp.fr)
Rosette Lidereau (r.lidereau@stcloud-huguenin.org)
Ivan Bieche (i.bieche@stcloud-huguenin.org)

Version: 4 Date: 21 January 2008

Author's response to reviews:

Please find enclosed the revised version of our manuscript "NF-kappa B genes have a major role in Inflammatory Breast Cancer".

Changes asked by the Biomed Central Editorial Production Team have been done in this version.

We hope that you will find it acceptable for publication in BMC Cancer and look forward to hearing from you.

Sincerely yours,
Dr Florence Lerebours